Author's response to reviews

Title: The relationship between spasticity in young children (18 months of age) with cerebral palsy and their gross motor function development

Authors:

Jan Willem Gorter (gorter@mcmaster.ca)
Olaf Verschuren (o.verschuren@dehoogstraat.nl)
Laura Van Riel (laura_van_riel@hotmail.com)
Marjolijn Ketelaar (m.ketelaar@dehoogstraat.nl)

Version: 6 Date: 31 August 2009

Author's response to reviews:

I changed a few typo's in the manuscript and also edited the references according to an example of an published article in BMC musculoskeletal disorders.